



Article Evaluation of a Resilience-Driven Operational Concept to Manage Drone Intrusions in Airports

Domenico Pascarella ^{1,*}, Gabriella Gigante ¹, Angela Vozella ¹, Maurizio Sodano ², Marco Ippolito ², Pierre Bieber ³, Thomas Dubot ³ and Edgar Martinavarro ⁴

- ¹ Safety and Security Department, CIRA—Italian Aerospace Research Centre, Via Maiorise, 81043 Capua, Italy; g.gigante@cira.it (G.G.); a.vozella@cira.it (A.V.)
- ² Soul Software—Italian Software Factory, Viale S. Modestino 33, 83013 Mercogliano, Italy;
- maurizio.sodano@soulsoftware.eu (M.S.); marco.ippolito@soulsoftware.eu (M.I.) ³ ONERA/DTIS, Université de Toulouse, BP 74025, 31055 Toulouse Cedex, France; pierre.bieber@onera.fr (P.B.);
- ONERA/DTIS, Universite de Toulouse, BP /4025, 31055 Toulouse Cedex, France; pierre.bieber@onera.fr (P.B. thomas.dubot@onera.fr (T.D.)
 Unmanned Agrial Platforms Department INITA National Institute of Agrospace Technology.
- ⁴ Unmanned Aerial Platforms Department, INTA—National Institute of Aerospace Technology, 28850 Torrejón de Ardoz, Spain; martinavarro@inta.es
- * Correspondence: d.pascarella@cira.it

Abstract: The drone market's growth poses a serious threat to the negligent, illicit, or non-cooperative use of drones, especially in airports and their surroundings. Effective protection of an airport against drone intrusions should guarantee mandatory safety levels but should also rely on a resilience-driven operational concept aimed at managing the intrusions without necessarily implying the closure of the airport. The concept faces both safety-related and security-related threats and is based on the definitions of: (i) new roles and responsibilities; (ii) a set of operational phases, accomplished by means of specific technological building blocks; (iii) a new operational procedure blending smoothly with existing aerodrome procedures in place. The paper investigates the evaluation of such a resilience-driven operational concept tailored to drone-intrusion features, airport features, and current operations. The proposed concept was evaluated by applying it to a concrete case study related to Milan Malpensa Airport. The evaluation was carried out by real-time simulations and event tree analysis, exploiting the implementation of specific simulation tools and the assessment of resilience-oriented metrics. The achieved results show the effectiveness of the proposed operational concept and elicit further requirements for future counter-drone systems in airports.

Keywords: drone intrusion; airport operations; resilience assessment; real-time simulation; event tree analysis; counter-drone systems

1. Introduction

Airports have become critical transportation infrastructures, representing the key nodes of the aviation system network with strict interdependencies between them, vehicles, and people. Operational issues at a single airport may impact the entire National Airspace System (NAS) and may trigger cross-border or worldwide ripple effects. The closedown of a large or medium-sized airport reverberates across the NAS, resulting in: (i) the immediate rerouting of hundreds of arrivals; (ii) the delay or the cancellation of thousands of domestic and international flights; (iii) cascading traffic disruptions, possibly affecting not only the specific day but also the course of the following days.

Tackling these airport issues requires overcoming traditional performance-driven approaches for the design of new solutions within air traffic management (ATM). Indeed, such approaches always apply the safety-as-a-priority paradigm by mainly focusing on system failures causing catastrophic or major accidents, which are events with large safety impacts. More generally, beyond these events, other ATM adverse events may push the dynamics of ATM far away from its point of operation (even without safety impacts)



Citation: Pascarella, D.; Gigante, G.; Vozella, A.; Sodano, M.; Ippolito, M.; Bieber, P.; Dubot, T.; Martinavarro, E. Evaluation of a Resilience-Driven Operational Concept to Manage Drone Intrusions in Airports. *Information* **2023**, *14*, 239. https:// doi.org/10.3390/info14040239

Academic Editors: Emil Pricop, Grigore Stamatescu and Jaouhar Fattahi

Received: 30 January 2023 Revised: 7 April 2023 Accepted: 9 April 2023 Published: 13 April 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). and dramatically affect the performance and economic aspects of the system [1]. The solutions to these adverse events have to consider the system as a whole, involving all of its dimensions, i.e., hardware, software, human, and environment. Accordingly, the systematic implementation of these solutions is being addressed by introducing the resilience concept into the ATM domain.

EUROCONTROL defines resilience as "the intrinsic ability of a system to adjust its functioning prior to, during, or following changes and disturbances, so that it can sustain required operations under both expected and unexpected conditions" [2]. This definition is mainly based on the work of Hollnagel, who has also generally addressed resilience as "the ability of a system or an organization to react to and recover from disturbances, a resilient ATM system responds to a reference disturbance within a limited time horizon by transient perturbation, i.e., by modifying its functioning [4].

Recent episodes highlight the increasing demand for resilient airports to ensure that the NAS is no longer vulnerable to disruptive events that occur at a single airport [5]. As an example, one may mention the failure of the electrical system at the Hartsfield-Jackson Atlanta International Airport in 2017, which shut down the airport for several hours and caused a loss of upwards of USD 50 million in revenue for the tenant airlines [5]. The resilience of an airport to an adverse event should exploit three phases [5]: (i) the "before" phase, for "resisting" through prior prevention and mitigation planning; (ii) the "during" phase, for "absorbing" through prior response planning; (iii) the "after" phase, for adapting through prior recovery planning. Clearly, proper implementation of these phases involves different dimensions of the planning process, i.e., technological, organizational, social, and economic.

However, in the last five years, the explosion of the drone market has brought out another serious threat against the resilience of airports related to the negligent, illicit, or non-cooperative use of drones. One of the well-known examples is represented by the drone intrusion at London Gatwick Airport in December 2018, during which an unknown number of overflying drones caused a 33 h paralysis of airport operations [6]. For this Gatwick episode, the European Union Aviation Safety Agency (EASA) has estimated a cost of EUR 64 million [7].

Moreover, EASA has published a counter-drone action plan [8] and some guidelines [7,9,10] in which counter-drone systems are recommended to mitigate drone intrusion impact on the airport ecosystem. These systems are also named counter-UAS or C-UAS systems, where the acronym UAS stands for unmanned aerial system. Thus, several C-UAS systems are being developed to satisfy the growing need to defend against intruding drones, especially at low altitudes and in the operational envelope of drones. Nevertheless, in the case of airport operations, these systems usually exploit a policy based on reactive countermeasures, consisting of the detection of the intrusion and of the subsequent closure of the overall airport until the threat has reduced. On the one hand, such a policy guarantees the safety levels of the airports in case of drone intrusions and is in line with the safety-asa-priority paradigm; on the other hand, it significantly penalizes the other performance levels of the airports since it implies relevant disruptions for the overall traffic network. In other words, airport drone intrusions are a glaring example of how the decoupled usage of safety aspects and of the other performance aspects may lead to an ultra-safe ATM, whose safety would conflict with the other performance dimensions (e.g., capacity, economy, time performance, human performance, etc.).

The only way to protect the resilience of the airports against drone intrusions is to build an effective drone intrusion management system (DIMS), which should deploy the three phases (before, during, and after) to allow the airport to resist, absorb, and adapt against the intrusion. For this purpose, in addition to reactive countermeasures, the DIMS should exploit both preventive countermeasures and proactive countermeasures. For example, among the preventive countermeasures, there are the deterrence and the denial to enter protected areas. Instead, proactive countermeasures are mainly based on: (i) creating a complete and timely situational awareness about drone intrusions in order to support the dynamic assessment of their risks; (ii) designing procedures and protocols to manage the intrusions in order to mitigate their impact as much as possible.

The ASPRID (Airport System PRotection from Intruding Drones) project [11,12] is an exploratory research project, funded by the Single European Sky ATM Research (SESAR) program, to develop an operational concept for the protection of airports against drone intrusions in a resilient way. The operational concept is based on proactive countermeasures and aims at increasing the resilience of the airport (that is, minimizing the performance degradation) against drone intrusions by avoiding the closure of the overall airport if not strictly necessary. This results in: (i) limiting the interruption to those operations that are strictly affected by the intruder; (ii) enabling "chirurgical" actions for mitigation and response, considering the closure of the airport as the last option. According to the phases and the dimensions for airport resilience in [5], the operational concept was designed with reference to:

- The "before" and "during" phases for "resisting" and "absorbing" the intrusion, in order to increase the resilience of the airport by means of mitigation and response planning;
- Technological and organizational dimensions in order to address a process model and an operational procedure for a proactive DIMS, employing different building blocks, from the detection up to the mitigation systems.

Our previous work in [13] puts forward the basis for a systematic process of risk management within the DIMS of an airport, describing a quantitative assessment of the historical features of drone intrusions in airports. Our other work in [14] describes a unified methodological framework for the risk assessment of airport drone intrusions, combining model-based and data-driven specifications. Leveraging the risk assessment results in [13, 14], this paper investigates the evaluation of the proposed resilience-driven operational concept for the DIMS, tailored on drone-intrusion features, airport features, and current operations. The concept was evaluated by applying it to a concrete case study related to Milan Malpensa Airport, in line with the case study described in [14]. The evaluation was carried out by real-time simulations and event tree analysis (ETA), exploiting the implementation of a specific simulation platform and the assessment of resilience-oriented metrics. The achieved results show the effectiveness of the proposed operational concept and elicit further requirements for future counter-drone systems in airports.

The article is organized as follows. Section 1 presents a brief overview of the issues for airport resilience related to drone intrusions, highlighting DIMSs as a proper tool for resilient protection. Section 2 analyzes the related work for both airport resilience and counter-drone systems in airports. Section 3 describes the proposed operational concept. Section 4 reports the methodology for the resilient-driven evaluation of the operational concept, including the description of the case study. Section 5 illustrates the evaluation results. Section 6 provides a discussion of the proposed operational concept, based on the achieved results. Section 7 provides the conclusions. Appendix A reports the detailed results of the simulation activities.

2. Related Work

This section describes the related work about airport resilience and counter-drone systems for airports.

2.1. Airport Resilience

In transportation networks, research on resilience mainly faces natural hazards or extreme weather events [15]. For the specific case of the air transport system, research on resilience mostly focuses on analyzing the complex systems within ATM and air traffic networks by means of complexity science in order to model the resilience of the system or network and to identify the most critical elements [16].

Reference [15] lists some European projects that have provided systematic approaches for the quantitative assessment of resilience and for a resilience-driven development of ATM solutions. For example, one mentioned project is Resilience2050 [17], which has developed models to support mathematical analyses of resilience in ATM scenarios, both in normal operations as well as disturbances [18]. The project considered disturbances related to weather conditions, bad visibility issues, runway configuration changes, and staffing problems [19].

Another mentioned project is the SAFECORAM (Sharing of Authority in Failure/ Emergency COndition for Resilience of Air traffic Management) project, which identified a model to evaluate resilience in terms of performance degradation in failure/emergency conditions and developed a concept to optimize the resilience index of an ATM system by means of task allocation and authority sharing between humans and automated components [20]. The concept was tested in some emergency scenarios, such as the aircraft's uplink loss during the en-route phase [21].

Thus, our proposed work represents the first attempt at explicitly evaluating airport resilience for threat scenarios related to drone intrusions.

2.2. Counter-Drone Systems for Airports

This subsection discusses the counter-drone systems for airports by describing both the current regulations and the research works.

The regulation of drone intrusions in airport environments has become a pressing issue in recent years, as the number of drones in operation continues to grow and the potential risks they pose to aviation safety become increasingly clear. In order to address this issue, several initiatives have been undertaken by key regulatory and standardization bodies, such as EASA and the European Organisation for Civil Aviation Equipment (EUROCAE), and industry associations, such as Airports Council International (ACI).

EASA formed the counter-UAS task force in November 2019 to work on objective 2 of the EASA C-UAS action plan [8], which is preparing aerodromes to mitigate risks from unauthorized drone use. The task force delivered a manual, named Drone Incident Management Manual for Aerodromes, which provides guidance on the management of drone incidents in and around airport environments. This manual is divided into three parts: Part 1 [7], which is publicly accessible, provides an overview of the challenges posed by unauthorized drones in airport environments; Part 2 [9] and Part 3 [10], which are more restricted in distribution due to their sensitive nature, provide guidance and recommendations for incident management and practical tools for implementation.

Within the same timeframe, EUROCAE and the Radio Technical Commission for Aeronautics (RTCA) jointly developed an Operational Services and Environment Definition (OSED), named ED-286 [22]. This document provides a detailed description of the operational services of a C-UAS system and the environment in which such a system will operate. It proposes operational requirements and associated assumptions and defines the following capabilities: global detection, decision support, neutralization, and data collection.

Finally, ACI formed a drone task force in February 2019 with the goal of discussing drone operations at airports with all relevant stakeholders, not only to exchange information but also to provide guidance on how drone operations could be facilitated whilst ensuring the necessary safety and security levels at airports. The task force created the Concept of Operations (ConOps) [23], which specifies a set of recommendations for the airport community (airport operators, air control, surveillance units, drone operators, etc.).

All these initiatives were created to provide a comprehensive framework for the management of drone incidents in airport environments. Together they represent a significant step forward in the regulation of drone intrusions in airport environments and helped to ensure the safety and security of passengers and personnel. Nevertheless, they require local implementations at the level of member states, depending on national regulatory specificities.

Concerning research works, reference [24] studied drone incidents in the vicinity of worldwide airports to deliver quantitative and qualitative analyses, proposing possible mitigation measures. Reference [25] describes an action plan for airport stakeholders to

defend an airport against misused drones. Lastly, several works provide a detailed analysis of the technological options for counter-drone systems, e.g., [25–27].

Leveraging the available regulations and C-UAS technological options, our proposed work also represents the first attempt at defining an operational concept for airport protection against drone intrusion, specifically designed to manage the resilience of airport operations.

2.3. Research Gaps and Proposed Innovation

This subsection provides a summary of the related work by pointing out the research gaps. In addition, it highlights the proposed innovation of this work.

In regard to airport resilience (Section 2.1), previous research was mainly related to the quantitative assessment of resilience against specific disturbances and off-nominal ATM scenarios, such as those related to weather conditions, bad visibility issues, runway configuration changes, staffing problems, and failures of the overall transportation system. Several works have developed resilience metrics and models by applying complexity science. To the best of our knowledge, no prior work has defined and assessed airport resilience for threat scenarios related to drone intrusions.

In regard to counter-drone systems for airports (Section 2.2), both regulations and research works have to be considered. The former provides a comprehensive framework for the management of drone incidents in airport environments. The latter mainly addresses safety analyses and deals with detailed technological options for "isolated" capabilities of counter-drone systems, e.g., in terms of detection and neutralization. To the best of our knowledge, no prior work has: (i) developed an "integrated" concept for airport protection against drone intrusions; (ii) considered resilience features for the design of counter-drone operations.

Based on the aforementioned research gaps, this work proposes a twofold innovation. On the one hand, it assesses the airport's resilience for threat scenarios related to drone intrusions by introducing explicit resilience metrics and by defining a resilience-driven evaluation methodology of counter-drone solutions. On the other hand, it designs an integrated and resilience-driven operational concept for drone intrusion management in airports by: (i) leveraging the available regulations and C-UAS technological options; (ii) managing both accidental and malicious intrusions; (iii) minimizing the airport's performance degradation.

3. Operational Concept for Drone Intrusion Management

This section describes ASPRID's operational concept, i.e., the proposed operational concept for the DIMS. Note that the operational concept "is designed to give an overall picture of the operations using one or more specific systems, or a set of related systems, in the organization's operational environment from the users' and operators' perspective" [28]. Thus, in our case, the ASPRID operational concept is a user-oriented and operator-oriented description of the operations and the building blocks that implement the DIMS. Such a description includes:

- ASPRID's process model, which specifies the actors and the workflow of the operations performed by the actors to accomplish the management of drone intrusions;
- ASPRID's operational procedure, which specifies the step-by-step instructions for the
 operators to accomplish the management of drone intrusions.

The operational concept considers both safety-related (i.e., accidental) and securityrelated (i.e., malicious) intrusions. In the case of security-related intrusions, these represent deliberate attacks, which may include both drone physical intrusions and drone cyber intrusions (i.e., cyberattacks using drones as cyber weapons) [14]. However, for the purposes of the ASPRID operational concept, only physical intrusions by single drones (i.e., not in teams or swarms) are considered as a reference threat in the stated problem.

The following subsections report the ASPRID process model and operational procedure.

3.1. Process Model

The ASPRID process model relies on the implementation of the roles and responsibilities of the actors in the Drone Incident Management Cell (DIMC), which refers to a group that responds to unauthorized drone activity in the airport environment, as recommended by EASA guidelines [7]. Main roles, such as air traffic controllers (ATCOs), including both tower controllers and ground controllers, or law enforcement authorities (LEAs), are refined to consider the management of drone intrusions according to: (i) regulation principles; (ii) the inclusion of the new actor for ASPRID's operational concept, i.e., the ASPRID system.

In detail, the ASPRID system is a technological actor performing the following functions:

- 1. The detection and tracking of unauthorized drones;
- 2. Alerting to drone intrusions;
- The classification and identification of unauthorized drones, i.e., reconnaissance of drone features (type, speed, etc.);
- 4. Providing support to threat assessments, i.e., the estimation of the level of risk related to drone intrusion (e.g., as a function of the type and the speed of the drone, of the current ongoing operations, etc.);
- 5. Providing support to C-UAS actions, i.e., the neutralization of unauthorized drones.

Thus, the ASPRID system provides the human actors with full situational awareness of the intrusion. Note that the first three functions are autonomous, whereas the last two are intended as support for the decisions of a human actor. Clearly, the required equipment for these functions is a design choice depending on the specific airport (some preliminary information is available for the simulation of the case study in Section 4.2.3). Moreover, the ASPRID system introduces a new human actor, named the ASPRID operator, who is responsible for supervising and controlling the ASPRID system.

The process model clarifies how the ASPRID system could support each DIMC subprocess and the associated communication exchanges. Three different phases may be distinguished:

- The strategic phase, during which roles and responsibilities are defined, tools are trained and configured, and main scenarios are simulated;
- The tactical phase, during which all actors manage the threat according to procedures defined in the strategic phase, the latest information data, and the ASPRID analysis;
- The post-operations (post-ops) phase, which is used to investigate past incidents and how processes may be improved.

Within the context of major airports, the ASPRID system should provide 24/7 situational awareness to all users by uninterrupted monitoring of the airport environment, making the tactical phase a rolling process that is always active. Figure 1 shows the flowchart for the main subprocesses of the tactical phase, using the Business Process Model and Notation (BPMN) [29]. It is divided into four swimlanes, representing the following actors or responsibilities: the DIMC, the ASPRID system (including ASPRID human roles that are detailed in Section 3.2), aircraft operators in the jurisdiction of the airport or air traffic control (ATC), and other external actors such as U-space. The DIMC lane itself is subdivided into three sub-lanes to distinguish actions realized within the DIMC by the aerodrome's safety and security operators, ATCOs and LEA. Boxes inside lanes represent activities under the responsibility of each actor, and plain arrows indicate the sequence of actions. Finally, dotted lines represent information exchanges between actors.



Figure 1. BPMN process diagram of the tactical phase of ASPRID process model.

3.2. Operational Procedure

ASPRID's operational procedure relies on the definition of critical assets of the airports. In detail, a critical asset (or critical element or, simply, asset) is any airport asset that exhibits a weakness against drone intrusions, i.e., the intrusion may result in a temporary or permanent interruption of the airport's operations involving the asset [14]. Thus, critical assets are those that are susceptible to mishap risks related to drone intrusions and are identified by means of a threat analysis within the risk assessment [14], which is part of the strategic phase of the ASPRID process model, as addressed in Section 3.1. Note that the assets may be both stationary and mobile (e.g., an aircraft).

For each asset, the threat analysis also defines a safety radius, which represents the minimum physical distance between the asset itself and any intruding drone in order for the asset to continue to operate under nominal conditions without interruption. Thus, the safety radius is intended as a safety bubble around the asset with different geometrical shapes, not necessarily circular. For example, runways are better associated with polygonal safety bubbles.

Figure 2 shows the principles for the operational management of a generic critical asset. Each asset is associated with the following alert levels in the case of drone intrusion:

- White level—The drone is outside of the safety radius. The asset operates in nominal mode.
- Yellow level—The drone is within the safety radius but does not physically interfere with the asset's operations. The asset operates in nominal mode but requires a high monitoring level.
- Orange level—The drone is within the safety radius and physically interferes with the asset's operations, but the collision between the drone and the asset is not imminent. The asset operates in degraded mode. Interdependent assets may also operate in degraded mode.
- Red level—The drone is within the safety radius, it physically interferes with the asset's operations, and the collision between the drone and the asset is imminent. The asset operates in suspended mode, i.e., it requires a shutdown for its operations. Interdependent assets or the full airport may also require a shutdown.



Figure 2. Alert levels and operational modes of a critical asset.

Note that the classification of the alert levels is strictly related to some configuration parameters, i.e., the sizes of the alert zones (yellow, orange, and red) and the functions available in the degraded operational mode. Such parameters are selected during the strategic phase of the ASPRID process model and depend on a number of features of the asset, such as its criticality to airport operations, its relationships with other assets, whether it is manned or not, whether the airport contains a redundant element or not, the presence of people around, etc. Moreover, this work assumes that the safety bubble and the alert levels of each asset are stationary and do not change with respect to the features of the intruder drone.

Figure 3 reports a tabular view of ASPRID's operational procedure, which is based on the aforementioned alert classification. The procedure addresses the actions to be performed by ASPRID operators and controllers (i.e., ATCOs), including tower controllers, ground controllers, and controllers for the terminal maneuvering area (TMA). These actions are triggered by specific conditions, which are mainly related to the alert levels in Figure 2. Only one condition is associated with the availability of a C-UAS action to neutralize an intruder drone: in this case, the ASPRID operator proceeds to issue the neutralization command only if authorized by the ATCO, who is responsible for checking that there are no moving aircraft near the target drone in order to avoid collateral damages. These guidelines are graphically specified in Figures 4 and 5, which, respectively, illustrate the BPMN process diagrams of the main process and of the infringement management subprocess within the ASPRID operational procedure. Note that, in Figure 4, the C-UAS management subprocess also includes LEAs; in Figure 5, the controller swimlane refers to ATCOs. Moreover, we assume that the main BPMN process (Figure 4) implies periodic detection and tracking for the ASPRID system; thus, the process starts on recurring moments in time by means of a timer start event.

Triggering Condition	ASPRID Operator Action	Controller Action
White infringement of unauthorized drone	 Monitor the object (periodic monitoring) (2) Keep the controller updated about the evolution of object's trajectory (significant updates) 	No actions
Yellow infringement of unauthorized drone	Inform the controller Monitor the object (continuous monitoring) Keep the controller updated about drone's trajectory and features, e.g., speed, model, etc. (continuous update)	 Move/redirect aircraft to keep them far from the infringed zone (e.g., use another runway for taxiing or take-off)
Orange infringement of unauthorized drone	As above	$(\widehat{1})$ Stop aircraft in nearby zones
Red infringement of unauthorized drone	As above	(1) Stop all airport operations
Available countermeasure to neutralize an unauthorized drone	 Ask the controller to confirm the neutralization action If the controller confirms the neutralization action, issue neutralization command Inform the controller about the neutralization result Repeat in case of failed neutralization 	 Check that there are no moving aircraft close to the neutralization area (close to the drone) Inform ASPRID operator about the previous check

Figure 3. Tabular view of ASPRID operational procedure.



Figure 4. BPMN process diagram of ASPRID operational procedure: main process.



Figure 5. BPMN process diagram of ASPRID operational procedure: subprocess for infringement management.

The designed procedure stops airport operations only if a red-zone infringement occurs. Moreover, the procedure also implicitly considers aircraft restoring: in case of a downgrade of the alert level (e.g., in case of an unauthorized drone's transition from orange to yellow infringement or from red to orange infringement), the controller performs the specific actions for the new alert level, which possibly implies a restoring of some aircraft previously stopped, according to their position with respect to the infringement zone. Instead, the procedure does not currently consider the restoration times due to the possible removal of the neutralized drone within the airport's perimeter.

4. Evaluation Methodology

This section describes the methodology for the resilience-driven evaluation of AS-PRID's operational concept. It discusses the general approach and the simulation strategy.

4.1. Approach

A specific evaluation methodology is applied to assess the effectiveness of the AS-PRID operational concept for the protection of airport operations against drone intrusions. Considering the resilience-driven nature of the concept, such effectiveness is intended as a positive impact with regard to the availability of the DIMS and is demonstrated by assessing the main performance benefits related to the increase in the airport's resilience to drone intrusions by means of the DIMS itself.

A concrete case study is arranged in order to provide tangible results about the performance benefits. In detail, the Milan Malpensa Airport is chosen as the reference airport for the case study to evaluate the proposed operational concept. The ICAO (International Civil Aviation Organization) airport code is LIMC. This is a proper reference airport for this evaluation because it is of medium complexity in the context of European airports. The airport is located 49 km from central Milan and has two passenger terminals as well as a dedicated cargo terminal. It presents two runways in a parallel configuration with various taxiways connecting these with the aforementioned terminals and other airside areas, as shown in Figure 6.



Figure 6. Runways and departure paths of Milan Malpensa Airport (LIMC).

The evaluation methodology combines: (i) a quantitative approach, based on real-time simulations; (ii) a qualitative approach, applying a risk analysis based on the achieved simulation results. The former approach involves the implementation of a proof-of-concept of the operational concept in a laboratory environment. Instead, the latter approach employs event trees.

Reference [14] provides a detailed description of the risk assessment of drone intrusions in airports, including event trees. The generic structure of such event trees is shown in Figure 7a, wherein the probability of a branch BProba is computed by using:

- Pts, the probability of the threat scenario;
- Pd, the probability of efficient drone detection;
- Pa, the probability of efficient drone assessment (i.e., identification);
- Pm, the probability of efficient drone mitigation (i.e., neutralization).



Figure 7. Evaluation approach. (**a**) Event tree for the assessment of threat scenarios [14]. (**b**) Evaluation workflow.

Instead, the proposed evaluation exploits real-time simulations to: (i) estimate resilience metrics and provide requirements for the implementation of the ASPRID system (C-UAS requirements); (ii) provide feedback to the risk assessment (ETA) in regard to safety aspects. The evaluation workflow is shown in Figure 7b. The following subsections provide further details about the approach for real-time simulations.

4.2. Real-Time Simulations

This section describes the simulation approach for the proposed evaluation. In detail, the approach uses real-time simulations, which means that some entities (actors or technological blocks) of the reference system are simulated by means of computer models, whereas other entities are played by human operators, requiring real-time constraints for simulation [30]. This approach is useful for evaluations that consider the behavior of human operators (e.g., pilots and controllers) and to validate the main functionalities of the reference system.

Moreover, given the functional scope of the ASPRID operational concept, the real-time simulations are intended as functional simulations in order to simulate the functional behavior as addressed by the specifications of the process model and the operational procedure. Thus, simulations represent "what" the ASPRID system shall do, not "how", by modeling its functional behavior. This allows for verifying the logical correctness of the concept and measuring its resilience-related benefits and impacts coherently from a functional viewpoint.

The following subsections point out: (i) the simulation platform; (ii) the simulation scenarios, including the operational environment and the threat scenarios; (iii) the performance metrics. The simulation platform is a set of interconnected tools and apps to simulate both the airport environment and the ASPRID operational concept. The platform is able to replicate the expected behavior of the ASPRID system, providing its functional simulation. Thus, the platform represents a software proof-of-concept of the operational concept in order to collect quantitative measures about its functional behavior.

Figure 8 illustrates the architecture of the simulation platform, which is composed of the following tools and apps [31]:

- ASPRID server—This is a backend that simulates the ASPRID system. It receives
 the configuration of the ASPRID system, the drone tracks, the traffic tracks, and the
 countermeasure commands. It estimates the state of the airport related to the threat,
 delivering the outputs of the ASPRID system, i.e., the information about the detection
 and tracking, identification, and neutralization of the intruder. It was developed in
 TypeScript and it runs on the Node.js runtime environment.
- ASPRID configurator—This is the tool in charge of the configuration of the ASPRID system. It allows: (i) the tagging of the critical assets in the available airport and specification of the related alert zones; (ii) the locating of the sensing and countermeasure units and specification of their characteristics. It has been developed in TypeScript.
- eArts (en-route aerodrome real-time simulator)—This is a real-time simulator of aerodrome, en-route, and approach operations [32]. It can reproduce the controller working positions (CWPs) for the different roles of ATCOs and for aircraft pseudo-pilots (Figure 9). It was developed in the JAVA and Kotlin programming languages. Based on the outcomes of the ASPRID server, eArts CWPs are used to show: aircraft and drone tracks on the map; drone identification data; and alert information.
- ASPRID web app—This is a frontend for the interaction with different classes of users, such as (Figure 10): ASPRID's operator; a human viewer (described in Section 4.2.3); a drone operator (i.e., the user who selects the trajectory and the features of the intruder drone); an LEA operator; etc. The ASPRID operator uses this app to issue neutralization commands within a simulation run. The web app was developed with the Flutter/Dart language. A demo web app is publicly available to replay some simulation exercises [33].



Figure 8. Architecture of the simulation platform.



Figure 9. eArts simulator.



Figure 10. ASPRID web app.

Lastly, a specific ASPRID postprocessor collects the logs of the simulation runs and evaluates the performance metrics (described in Section 4.2.5). Due to the exploratory research carried out in the ASPRID project, the proposed evaluation does not consider the sensitivity of the results, even if the results should be able to be directly extended to all the scenarios (e.g., other airports) with the same traffic/operational environment.

4.2.2. Simulation Scenarios

Simulation scenarios are the representations of the environment and of the behaviors that are used for the real-time simulation of the proposed operational concept. The adopted simulation approach entails the following classes of simulation scenarios:

- The solution-scenario class, which is composed of simulation scenarios reproducing a given environment with the implementation of the operational improvement (i.e., the proposed operational concept) that is the subject of evaluation;
- The reference-scenario class, which is composed of simulation scenarios reproducing the same environment as a solution scenario without the implementation of the operational improvement that is the subject of evaluation.

Both classes are referred to within LIMC airport, coherently with the stated case study. Moreover, each simulation scenario is configured by defining its following components: (i) an airport environment, which in turn consists of the traffic environment and the operational environment for the selected airport; (ii) a threat scenario. The remainder of this subsection describes the traffic environment, whereas the operational environment and the threat scenarios are described, respectively, in Sections 4.2.3 and 4.2.4.

The traffic environment reproduces the air traffic distribution within the simulation scenario. All the solution scenarios and reference scenarios share the same traffic environment, which is built upon a real traffic distribution from a representative day at LIMC airport. A medium-traffic volume is used, which is between 50% and 75% of the high-traffic volume, and it includes only departing flights. In detail, all the simulation runs adopt the real planned departure traffic of LIMC for the day of 8 February 2019 at 15:50 GMT. The average time window used in the simulations is 40 min. The choice of using only departing flights is strictly related to the human team involved in the simulation runs, as further explained in Section 4.2.3. No other off-nominal events (in addition to drone intrusions) are included within the traffic environment.

4.2.3. Operational Environment

The operational environment reproduces the set of airport capabilities and procedures, including those related to counter-drone management operations. In detail, the following entities make up the simulation of the operational environment: (i) counter-drone sensing; (ii) counter-drone alerting; (iii) counter-drone neutralizing; (iv) human operators and counter-drone operational procedure.

Counter-drone sensing entities simulate the detection and tracking capabilities of the airport with respect to drone intrusions. Each sensor is modeled by means of the following parameters: spatial coverage, which is the geographic area covered by the sensor for the drone sensing; efficiency, which is the probability of successful sensing at each (simulated) time instant in case the drone is inside the covered area of the sensor.

For reference scenarios, counter-drone sensing consists only of human sightings, which are simulated by means of "artificial" sensors, which do not represent real physical sensors, but simulate the sighting of drones by humans. Human sightings are modeled only by stationary sensors and by properly configuring their parameters. The following classes of sensors are used for human sightings in reference scenarios:

- Artificial airport sensors—They are displaced within the airport perimeter. They also include an artificial tower sensor, related to ATCOs in the tower.
- Artificial external sensors—They are displaced outside the airport perimeter. They simulate the human sighting of drones from the outside (e.g., from aircraft pilots).

For solution scenarios, the following classes of sensors are used to simulate drone sensing technologies according to the ASPRID operational concept:

- Radar sensors—They simulate the detection-and-tracking capabilities by means of radars. They are able to detect and track the radar signature of an object, but they are not able to identify its nature. Thus, radars detect and track unknown objects, but they cannot confirm whether the object is a real unauthorized drone.
- Identification sensors—They simulate the detection-and-tracking capabilities by means
 of specific sensors (e.g., optical sensors, acoustic sensors, infrared sensors, etc.), which
 may also provide identification. They are able to: detect and track an object; confirm
 whether it is an unauthorized drone; recognize the model; provide an estimation

of speed, mass, and payload. This class of sensors is divided into the following subclasses: (i) airport identification sensors, which represent identification sensors displaced within the airport perimeter; (ii) external identification sensors, which represent identification sensors displaced outside the airport perimeter, simulating the identification of drones by means of external support, such as LEAs.

Table 1 reports the parameter values of the counter-drone sensing entities. For reference scenarios, the configuration of the spatial coverages and of the efficiency is based on qualitative criteria related to the line of sight and the density of people on the ground. For solution scenarios, the parameter values reflect the technological features of the sensors according to a specific technological analysis carried out within the ASPRID project.

Simulation Scenarios	Sensor Class	Spatial Coverage (m)	Efficiency
D (Artificial airport sensors (tower)	500	50%
Keterence	Artificial airport sensors (other) Artificial external sensors	400 600	50%
Solution	Radar sensors Identification sensors	3500 1000	25% 75%

Table 1. Parameter values of the counter-drone sensing entities.

Figures 11 and 12, respectively, show the displacement of counter-drone sensors within reference scenarios and solution scenarios. For reference scenarios, the displacement of artificial airport sensors qualitatively considers the density of human actors on the ground during the intrusion, favoring sensing near critical sites such as the taxiways, the parking area, etc. For solution scenarios, displacement is a design choice for the operational concept within the case study.



Figure 11. Displacement of counter-drone sensors for reference scenarios.



Figure 12. Displacement of counter-drone sensors for solution scenarios. (a) North part of LIMC. (b) South part of LIMC.

Counter-drone alerting entities simulate the alerting capabilities of the airport with respect to drone intrusions in the vicinity of a critical asset. These entities are present only in solution scenarios, and their configuration applies the following rules according to the ASPRID operational concept:

- 1. Each asset is associated with three alerting zones, i.e., yellow, orange, and red;
- 2. The risk levels of the alert are in ascending order from yellow to orange and then to red, and consider the distance of the drone from the asset;
- 3. The geometrical shape of an alerting zone may be any closed shape.

The covered stationary assets of LIMC are: runway, 35L; runway, 35R; approach area of runway, 35L (APP 35L); approach area of runway, 35R (APP 35R); departure area of runway, 35L (DEP 35L); departure area of runway, 35R (DEP 35R); tower; finger 1; finger 2; finger 3; fire station. Figure 13 highlights the stationary critical assets and illustrates the displacement of their alerting zones. In regard to the aircraft, the following extensions of the alerting zones are used: 700 m for the yellow zone; 100 m for the orange zone; 50 m for the red zone. With regard to the stationary critical assets, a size of 600 m is used for the yellow zones of all the assets. Different values are used for the orange and red zones, related to the effective dimension of the asset.



Figure 13. Displacement of counter-drone alerting zones of stationary assets for solution scenarios. (a) North part of LIMC. (b) South part of LIMC.

Counter-drone neutralization entities simulate the neutralization technologies and the related neutralizers or countermeasures. These entities are present only in solution scenarios. For the purposes of the proposed evaluation, jamming and net protection (i.e., physical interception) are simulated. Jammers, especially, are deemed more common with respect to spoofing; thus, they are part of the operational environment. Neutralizers are modeled by means of the following parameters: latitude and longitude of the position of the neutralizer; spatial coverage, i.e., the radius of the zone covered for the neutralization action; reaction time, i.e., the latency between the issuing of the neutralization command from the ASPRID operator and the execution of the neutralization action; success probability, referring to a single neutralization attempt. The following neutralizers are used:

- Airport jammers (internal to the airport);
- LEA jammers (external to the airport);
- Airport net.

Table 2 reports the parameter values of the counter-drone neutralization entities, which are selected according to a specific technological analysis carried out within the ASPRID project. Figure 14 shows the displacement of neutralizers that is adopted for the evaluation within the case study.

Table 2. Parameter values of the counter-drone sensing entities.

Neutralizer	Spatial Coverage (m)	Reaction Time (s)	Success Probability
Airport jammer	1100	1	65%
LEA jammer	1300	30	65%
Airport net	100	2	50%



Figure 14. Displacement of counter-drone neutralization entities for solution scenarios. (**a**) North part of LIMC. (**b**) South part of LIMC.

The counter-drone operational procedure represents the procedure to manage drone intrusions within the operational environment. For reference scenarios, the following operators are involved: (i) one person plays the "controller" role, which includes both the ground controller and the tower controller; (ii) one person plays the "human viewer" role for the sighting of unauthorized drones and for informing the controller in case of sightings. The following procedure is adopted:

1. In case of a drone sighting by a human viewer, this viewer informs the controller;

2. The controller stops all ongoing operations and does not allow new aircraft to taxi from those available at the gates.

Considering the average time window of the simulations (40 min), airport operations are never resumed after an unauthorized drone sighting. Moreover, in the case of a collision between the drone and a critical airport asset, the simulation run is stopped.

Instead, for solution scenarios, the following operators are involved: (i) one person plays the controller role, which includes both the ground controller and the tower controller; (ii) one person plays the "ASPRID operator" role. The counter-drone operational procedure is the one representing the ASPRID operational procedure, described in Section 3.2.

No pseudo-pilot is involved in the simulations of either reference scenarios or solution scenarios. Thus, it is assumed that the aircraft always executes the controller's instructions.

4.2.4. Threat Scenarios

Threat scenarios or hazard scenarios are descriptions of how a threat or hazard might materialize and represent a logical sequence from a hazard to its consequence [10]. A threat scenario specifies the chain of events or occurrences that take place, starting with one or more threats and ending with the consequences of an incident [34]. In this work, threats are unauthorized drone intrusions in the airport, and threat scenarios specify the chain of events or occurrences that:

- Start with a drone intrusion;
- Affect the operations of the airport and of its DIMS (if any);
- End with the consequences of an incident (i.e., collision with an airport's asset) or with a safe termination of the intrusion (e.g., the exit of the drone from the airport and its surroundings, neutralization of the drone, etc.).

For the purposes of the proposed resilience evaluation, an extended risk assessment process has to be adopted with respect to threat scenarios, in order to include both the resilience-driven and the simulation-based perspectives. Indeed, all the threat scenarios should be simulated within both reference scenarios and solution scenarios to: (i) quantify the level of risk for each threat scenario; (ii) quantify the level of the airport's resilience for each threat scenario; (iii) compare the results between reference and solution scenarios to provide a quantitative evaluation of the expected performance of the operational concept.

In general, threat scenarios should be assessed by considering the possible physical interference of an intruder drone with respect to a given airport's assets. In principle, such behavior may be traced to two main classes of drone interference: (i) fly-by, which refers to the proximal presence of a drone with respect to an asset or an airspace; (ii) collision, which refers to the crash of the drone with an asset, usually implying damage to the asset according to the kinetic energy of the drone. Several control variables may influence the evolution of a threat scenario and the outcomes of the evaluation, such as:

- The features of the intruder, e.g., in terms of trajectory, speed (also including constantspeed intrusion or variable-speed intrusion), drone type and related mass, etc.;
- The effectiveness of the sensing technologies of the DIMS, e.g., in terms of space coverage, accuracy, acquisition rate, success probability, dependency on meteorological conditions, etc.;
- The effectiveness of the neutralization technologies of the DIMS, e.g., in terms of technology (e.g., jamming, protection net, etc.), spatial coverage, latency, success probability, stationary/mobile type, dependency on meteorological conditions, collateral damages, etc.;
- The spatial displacement of the sensor and neutralization components;
- The type of airport traffic.

Thus, the evaluation of an operational concept for the DIMS is generally a complex problem due to the large number of threat scenarios, control variables, and their values. However, considering the exploratory nature of ASPRID research, the proposed evaluation does not assess the influence of the aforementioned variables on the outcomes, except for the bidimensional intrusion trajectory. Indeed, this is deemed an essential variable since it differentiates the main plausible classes of threat scenarios based on the airport's asset(s) that are affected by the intrusion trajectory, including standard instrument departure routes (SIDs) and standard arrival routes (STARs). The speed of the drones is simulated as a constant in the range of $5 \div 10 \text{ m/s}$, which reflects the most common speed of commercial drones. In regard to the intrusion's altitude, it is assumed to be below a threshold that would allow the intrusion to interfere with airport operations. Such constraints are related to the simulation tools, whereas the ASPRID operational concept is basically a 3D solution.

Table 3 lists the threat scenario classes that have been identified for LIMC and the instances of the classes (with the related figures) that have been simulated for the evaluation.

Threat Scenario Class	Description	Instances		
Threat on 35L	This class includes intrusion trajectories representing direct threats to the runway 35L.	Drone 01—It affects the north part of 35L (Figure 15) Drone 02—It affects the middle part of 35L (Figure 15) Drone 03—It affects the south part of 35L (Figure 15)		
Threat on 35R	This class includes intrusion trajectories representing direct threats to the runway 35R.	Drone 01—It affects the north part of 35R (Figure 15) Drone 02—It affects the middle part of 35R Figure 15) Drone 03—It affects the south part of 35R (Figure 15)		
Threat parallel to RWYs	This class includes intrusion trajectories representing direct threats running parallel to the runways.	Drone 01 and Drone 04—They affect 35L (Figure 16) Drone 02 and Drone 03—They affect 35R (Figure 16)		
Threat on fingers	This class includes intrusion trajectories representing direct threats to the fingers.	Drone 01 (Figure 17)		
Threat on tower	This class includes intrusion trajectories representing direct threats to the tower.	Drone 01 (Figure 17)		
Threat on routes	This class includes intrusion trajectories representing direct threats to the routes, i.e., for the SIDs and STARs of the airport.	 Drone 01—It affects the SIDs, without entering the airport perimeter (Figure 18) Drone 02—It affects the STARs, without entering the airport perimeter (Figure 18) Drone 03—It affects the SIDs and then enters the airport perimeter (Figure 18) Drone 04—It affects the STARs and then enters the airport perimeter (Figure 18) 		

Table 3. Threat scenario classes and related instances for the proposed evaluation.



Figure 15. Instances of threat scenario classes. (a) "Threat on 35L". (b) "Threat on 35R".



Figure 16. Instances of threat scenario class "Threat parallel to RWYs".



Figure 17. Instances of threat scenario classes. (a) "Threat on fingers". (b) "Threat on tower".



Figure 18. Instances of threat scenario classes. (a) "Threat on routes (SIDs)". (b) "Threat on routes (STARs)".

4.2.5. Performance Metrics

The basic idea of the proposed simulation-based evaluation consists of a quantitative comparison between the simulation results of reference scenarios and solution scenarios. This requires the setup of a specific performance framework, i.e., a proper set of performance metrics to allow the measurement of the effectiveness of the operational concept from a resilience-driven perspective.

The main reference for the definition of such a performance framework is represented by the SESAR performance framework [35], which reports: (i) the performance management process for measuring the performance and tracking the status of SESAR solutions with regard to performance targets; (ii) a set of key performance areas (KPAs), key performance indicators (KPIs), and performance indicators (PI). In addition, the SESAR performance framework specifies some focus areas within each KPA that are typically related to some identified performance issues of ATM. In detail, a resilience focus area is addressed within the capacity KPA in order to measure the ability [35]:

- To deliver performance goals during abnormal operating conditions by reducing impact and decreasing time to recover;
- To withstand and recover from planned and unplanned events and conditions that cause a loss of nominal capacity.

In this focus area, the resilience of an ATM system is influenced by [35]: (i) its anticipation and handling in regard to degraded conditions; (ii) its recovery from degraded to normal conditions. Planned and unplanned events and conditions include: weather, such as thunderstorms, strong winds, freezing conditions, and low visibility conditions; infrastructure degradation, such as technical failures, strikes, accidents, and runway maintenance.

However, the SESAR performance framework does not explicitly consider airports' counter-drone operations. Thus, within the proposed simulation-based evaluation, some specific performance metrics are defined to measure: (i) the anticipation and handling of airports in regard to degraded conditions due to drone intrusions; (ii) the recovery of airports from degraded to normal conditions in the case of drone intrusions. For this purpose, the following sets of metrics are introduced, based on the data provided by the simulation tools (as described in Section 4.2.1):

- Safety metrics (Table 4)—They measure the variation in safety margins for airport
 operations due to drone intrusions. They are defined in connection with the critical
 assets, the threat scenarios, and the operational procedure for drone intrusions. They
 basically consider the infringements of the intrusions with respect to the critical assets
 of the airports.
- Time performance metrics (Table 5)—They measure the variation in time performance for airport operations due to drone intrusions. They consider the punctuality of the airport traffic.
- Workload metrics (Table 6)—They measure the variation in human workload for airport operations due to drone intrusions. The human workload is related to controllers and is basically quantified in terms of the number of clearances.
- Capacity metrics (Table 7)—They measure the variation in airport capacity due to drone intrusions. Capacity is assessed by means of arrival and departure rates for each runway.

Table 4.	Safety	metrics.
----------	--------	----------

Safety Metric	Description
NYI	Number of yellow-zone infringements (NYI) of drone intrusions
NOI	Number of orange-zone infringements (NOI) of drone intrusions
NRI	Number of red-zone infringements (NRI) of drone intrusions
MTYI	Mean time of yellow-zone infringement (MTYI) of drone intrusions
MTOI	Mean time of orange-zone infringement (MTOI) of drone intrusions
MTRI	Mean time of red-zone infringement (MTRI) of drone intrusions
WTYI	Worst time of yellow-zone infringement (WTYI) of drone intrusions
WTOI	Worst time of orange-zone infringement (MTOI) of drone intrusions
WTRI	Worst time of red-zone infringement (MTRI) of drone intrusions
NYI-P	Number of yellow-zone infringements of drone intrusions against planes
NOI-P	Number of orange-zone infringements of drone intrusions against planes
NRI-P	Number of red-zone infringements of drone intrusions against planes
NYI-A	Number of yellow-zone infringements of drone intrusions against stationary assets
NOI-A	Number of orange-zone infringements of drone intrusions against stationary assets
NRI-A	Number of red-zone infringements of drone intrusions against stationary assets

Time Performance Metric Description Number of delayed arrival flights (NDAF) with a delay greater than a given threshold NDAF NDDF Number of delayed departing flights (NDDF) with a delay greater than a given threshold PDAF Percentage of delayed arrival flights (PDAF) with a delay greater than a given threshold PDDF Percentage of delayed departing flights (PDDF) with a delay greater than a given threshold Mean delay of arrival flights (MDAF) with a delay greater than a given threshold MDAF MDDF Mean delay of departing flights (MDDF) with a delay greater than a given threshold Worst delay of arrival flights (WDAF) with a delay greater than a given threshold WDAF WDDF Worst delay of departing flights (WDDF) with a delay greater than a given threshold

Table 5. Time performance metrics.

Table 6. Workload metrics.

Workload Metric	Description
NCAF	Number of controller clearances for arrival flights (NCAF)
NCDF	Number of controller clearances for departing flights (NCDF)
NNCAF	Normalized number of controller clearances for arrival flights (NNCAF)
NNCDF	Normalized number of controller clearances for departing flights (NCDF)

Table 7. Capacity metrics.

Capacity Metric	Description
AR	Arrival rate (AR) per 15 min for a given runway
DR	Departure rate (AR) per 15 min for a given runway

In addition to the aforementioned metric sets, some comprehensive performance degradation metrics are introduced to provide synthetic and quantitative indexes about the overall resilience, i.e., the performance degradation of airport operations due to drone intrusions. Every performance degradation metric consists of a classification scale with different degradation levels, each of which is associated with a triggering rule specifying the logical condition that underlies the corresponding level. The following are the proposed performance degradation metrics:

- Safety degradation metric (SDM, Table 8)—It includes four degradation levels (catastrophic, hazardous, major, minor, and negligible), which are related to the infringements of the assets by means of drone intrusions. The triggering rules in Table 8 do not consider: collisions between drones and assets, since red-zone infringements already trigger catastrophic (for aircraft) and hazardous (for stationary assets) degradations; durations and speed criticalities of the intruder trajectories in the infringement zones, since these are not used as control variables for the threat scenarios, as explained in Section 4.2.4.
- Time performance degradation metric (TDM, Table 9)—It includes three degradation levels (high, medium, and low), which are related to the mean delays for arrival and departure traffic.
- Workload degradation metric (WDM, Table 10)—It includes three degradation levels (high, medium, and low), which are related to the normalized numbers of controllers' clearances to manage arrival and departure flights.
- In the tables for the degradation metrics (Tables 8–10), the triggering rules are applied in an ordered way, starting from the ones applicable for greater degradation levels. For example, in Table 9 for TDM, the rule for the medium level is applicable if the rule for the high level is not triggered; the rule for the low level is applicable if the rules for the high and medium levels are not triggered.

The proposed metrics represent the possible indicators for a performance framework to evaluate the counter-drone behaviors of airports and their resilience against drone intrusions. Starting from these metrics, further specific indexes may be defined to evaluate additional resilience aspects (e.g., the percentage of diverted arrival traffic due to drone intrusions) or additional performance dimensions, such as the one related to the effectiveness of the counter-drone technology (e.g., the success rate of the overall detection system, the success rate of the overall neutralization system, etc.).

Table 8. Safety degradation metric.

Safety Degradation Level	Safety Degradation Value	Triggering Rule
Catastrophic	4	NRI - P > 0
Hazardous	3	(NOI - P > 0) OR (NRI - A > 0)
Major	2	(NYI - P > 0) OR (NOI - A > 0)
Minor	1	NYI - A > 0
Negligible	0	All other conditions

Table 9. Time performance degradation metric.

Time Performance Degradation Level	Time Performance Degradation Value	Triggering Rule
High	2	(MDAF > 20 min) OR (MDDF > 20 min)
Medium	1	$\begin{array}{l} (\text{MDAF} \ \in [10 \ \text{min}, \ 20 \ \text{min}]) \\ \text{OR} \\ (\text{MDDF} \ \in [10 \ \text{min}, \ 20 \ \text{min}]) \end{array}$
Low	0	$(MDAF < 20\ min)\ OR\ (MDDF < 20\ min)$

Table 10. Workload degradation metric.

Workload Degradation Level	Workload Degradation Value	Triggering Rule
High	2	(NNCAF > 10) OR (NNCDF > 10)
Medium	1	$(\text{NNCAF} \in [5, 10]) \text{ OR } (\text{NNCDF} \in [5, 10])$
Low	0	All other conditions

5. Results

This section describes the evaluation results, highlighting both the simulation and risk assessment results.

5.1. Simulation Results

In line with the approach described in Section 4, each threat scenario was simulated in order to estimate the performance metrics for both the reference scenarios and solution scenarios. The tables in Appendix A (Tables A1–A5) report the detailed results for each metric set (safety, time performance, workload, and capacity). Considering the sensing configuration for reference scenarios, some of the designed threat scenarios ("threat parallel to runways—Drone 02 and Drone 03", "threat on routes—Drone 02 and Drone 04") were deemed non-significant for the simulations since they did not imply a drone sighting. Thus, such threat scenarios were not included in the evaluation.

The following considerations may be made relative to the performance analysis of the simulation results:

- Safety (Tables A1 and A2)—For the sake of brevity, only a subset of the performance metrics has been reported in the tables.
 - a. Reference scenarios—All the threat scenarios lead to a catastrophic (4) value of safety degradation, as shown in Table A1. Several threat scenarios imply a collision between the intruder drone and a stationary asset, such as the scenarios "threat on 35L" and "threat on 35R".
 - b. Solution scenarios—No threat scenarios imply collisions between drones and airport assets and no catastrophic (4) values are recorded for the safety degradation, as shown in Table A2. The only situation with a hazardous (3) SDM is represented by the scenario "threat on fingers—Drone 01", due to an orange zone

infringement with a plane. However, looking at the trajectory of the intruder in Figure 17, this infringement regards only stationary planes at the fingers. Thus, solution scenarios limit the reduction of safety margins.

- Time performance (Table A3)—With respect to the metrics in Table 5, two metrics are added in Table A3, which are: the number of departures, i.e., the number of flights available at the gates and ready for taxiing, and the number of takeoffs. Note that the number of departures is generally greater in solution scenarios since reference scenarios stop departure operations in the case of a drone sighting. Moreover, the number of takeoffs is always less than the number of departures, even in solution scenarios. This is because, in the case of drone neutralization, recovery is simulated by just restoring the previously stopped aircraft without approving new aircraft for taxing from those available at the gates.
 - a. Reference scenarios—Almost all the threat scenarios lead to a high (2) time performance degradation. Only the scenarios "threat on 35R—Drone 01" and "threat parallel to runways—Drone 04" lead to medium (1) and low (0) values for TDM. This is due to the specific trajectory of the drone intrusion in these scenarios, which is such that it postpones the drone sighting and implies fewer delays. This occurs even if a collision is recorded for the scenario "threat on 35R—Drone 01" considering that the collision occurs "late" in the simulation timeframe.
 - b. Solution scenarios—Almost all the threat scenarios lead to a low (0) degradation of time performance, with the exception of the scenario "threat on routes—Drone 01". This scenario is especially challenging from the point of view of the operational procedure since the drone's trajectory (Figure 18) is such that no neutralization is possible. For the given scenarios and in the simulated time-frame (on average, 40 min), the ASPRID operational concept achieves a mean departing delay of about 400 s, which is less than a third of the mean departing delay of reference scenarios (about 1360 s). On the one hand, this gives a clear and quantitative idea of the time performance improvement in solution scenarios with respect to reference scenarios. On the other hand, this improvement is underestimated since several flights in reference scenarios would garner greater delays (or would even be canceled) with longer simulation timeframes.
- Workload (Table A4)—The number of departures is included also in the analysis of workload metrics, as it is for the time performance.
 - a. Reference scenarios—Almost all the threat scenarios lead to a low (0) degradation of workload. This is coherent with the idea that, in reference scenarios, the operational procedure simply stops all airport operations in the case of drone sightings without providing a significant burden to controllers' workload. However, this is also due to the fact that for these scenarios, the simulation never restores airport operations after an unauthorized drone sighting; in the case of a collision between the drone and a critical airport asset, the simulation is stopped without considering restoration.
 - b. Solution scenarios—Almost all the threat scenarios lead to a medium (1) degradation of workload. This is coherent with the greater awareness and the early management of drone intrusions and the new guidelines of the ASPRID operational procedure, which prescribe different controller actions according to the current alert level of the threat. However, only a slight workload increase is observed in solution scenarios. Indeed, the normalized number of clearances per departing flight is on average 5.14, which is only 0.3 greater with respect to reference scenarios (4.86).
- Capacity (Table A5)—The results for capacity provide an additional view with respect to time performance. First of all, the same observations may be made related to the number of flights released for departures. Moreover, a clear capacity improvement

may be observed. Indeed, solution scenarios show an average increment of about four flights/15 min for the departure rate with respect to reference scenarios.

By way of example, Figure 19 shows the performance results for the MDDF (belonging to the time performance set) and for the departure rate (belonging to the capacity set), also highlighting the mean values (red vertical lines) and the intervals for the standard deviation (yellow vertical lines). This figure gives a clear and quantitative idea of time performance and capacity improvements.



Figure 19. Simulation results. (a) Histograms of mean delay of departing flights for reference scenarios (up) and solution scenarios (down). (b) Histograms of departure rate (per 15 min) for reference scenarios (up) and solution scenarios (down). Red vertical lines represent the mean values, whereas yellow vertical lines represent the intervals for the standard deviation.

5.2. Risk Assessment Results

Our previous work [14] developed several detailed event trees for threat scenarios regarding the intrusion of a drone in the departure path of the runways at LIMC. Figure 20 shows two event trees for departures on runways 35R or 35L: the former for an intrusion taking place less than 1 km from the airport; the latter for an intrusion taking place at a greater distance.

The event trees highlight in pink the branches that represent an unacceptable risk. In particular, for the first event tree, the following two branches are unacceptable for the intrusion that occurred close to the airport: Branch 2 and Branch 6, which both involve the loss of efficiency of the mitigation (i.e., neutralization) capability. Instead, for the second event tree, the following three branches pose unacceptable risks: Branch 5, Branch 6, and Branch 8. They are all related to a loss of efficiency in the detection means. In addition to this condition, Branch 6 involves a loss of efficiency in the mitigation capability, whereas

5

6.5

7.0



Branch 8 involves a loss of efficiency in all the technical means. See [14] for further details about the estimation of the probabilities and the identification of the unacceptable branches.

Figure 20. Event trees for the intrusion of an unauthorized drone in LIMC 35 departure path when takeoff is occurring [14]. (a) <1 km from runway. (b) >1 km from runway.

Compared to these event trees, some preliminary considerations may be introduced for the ASPRID operational concept with regard to safety aspects. In detail, the following results have been achieved for the simulated solution scenarios:

- ASPRID's operational concept has always efficiently detected the intruder drone in the given threat scenarios. This contributes to a potential de-risking of all the unacceptable branches in Figure 20b.
- With regard to threat assessments, no consideration may be provided since the successful/unsuccessful assessments of intruder drones have not been modeled in the real-time simulator.
- ASPRID's operational concept has always neutralized the intruder drone in the given threat scenarios, where allowed by the countermeasures' coverage. In case a neutralization was not available due to the intruder's trajectory (i.e., in the scenarios "threat parallel to runways" and "threat on routes—Drone 01"), the ASPRID operational procedure ruled out safety issues. This is confirmed by the results in Table A2, which show that, with regard to the abovementioned scenarios: (i) no collisions and no red alerts occurred; (ii) the safety degradation was only major (2). This contributes to a potential de-risking of Branch 2 and Branch 6 in Figure 20a.

Clearly, more extensive simulations are required to confirm these preliminary considerations about the safety impacts of the ASPRID operational concept and its de-risking effect for unacceptable branches in ETA. Moreover, such considerations are valid in the case of the operational environment used for the simulation of solution scenarios, which may be interpreted as a set of requirements in order to reach the estimated values for the performance metrics. In other words, the proposed approach enables reverse reasoning to allocate the required reliability to the technical means for detection, assessment, and neutralization.

6. Discussion

This section provides a discussion of the achieved results for the proposed case study. Such results demonstrate the resilience-driven effectiveness of the ASPRID operational concept to protect airport operations against drone intrusions. Indeed, the proposed evaluation methodology has assessed the main performance benefits related to the increase in the airport's resilience to drone intrusions by means of the ASPRID concept. These benefits may be summarized in the following key points in the context of drone intrusions:

 Guarantee the safe continuity of airport operations in normal or degraded modes—The safety of airport operations is preserved, as demonstrated by the simulation results for the safety metrics. Moreover, the ASPRID system potentially contributes to the de-risking of most critical branches in the event trees, as preliminarily discussed in regard to risk assessment results. These numbers are reached despite that the closure of the overall airport was avoided in all the solution scenarios by means of the ASPRID operational procedure, which limited the interruption to those operations that are strictly affected by the intruder.

- Reduce time performance and capacity degradations—This benefit is demonstrated by the simulation results for time performance and capacity metrics, which report a clear improvement with respect to reference scenarios. Clearly, such a benefit directly implies an increase in airport resilience in the face of drone intrusions.
- Limit the increase in workload—An increase in the human workload is observed for solution scenarios, as expected. Anyway, even if this does not really represent a performance benefit, the workload increase is slight with respect to reference scenarios. Additional mitigation of this increase may be further explored in future works.
- Improve the situational awareness and early-management capabilities of airport operations with respect to intrusions—This benefit is not directly measured, but it is tangible due to the alerting functions. Moreover, the previous benefits are a clear consequence of this one.
- Lastly, it is worth noting that, with respect to reference scenarios, the assessed improvement in solution scenarios is even greater in the case of long-lasting drone intrusions.

7. Conclusions

This paper deals with the protection of airport operations against drone intrusions. It proposes a resilience-driven operational concept for such protection in order to minimize the performance degradation of airport operations by avoiding the closure of the overall airport if not strictly necessary. The operational concept is modeled in terms of a process model and an operational procedure. Moreover, a resilience-driven evaluation is proposed to assess the performance benefits of the concept. The evaluation exploits a combination of real-time simulations and risk analysis activities on event trees. The proposed evaluation is applied to a concrete case study related to the Milan Malpensa Airport.

Several research directions will be considered for future work. Firstly, the evaluation may be extended by also exploiting fast-time simulation tools, in order to perform the extensive simulations that are required to study the impact of all the control variables on the concept's effectiveness for managing threat scenarios. Then, real-time simulations may be enriched with simulation campaigns involving several controllers and experts (e.g., representing LEAs, aerodrome safety and security operators, etc.), in order to learn more about the sensitivity of real-time simulation results with respect to real users. In detail, a simulation-based gaming exercise has already been performed [36] and will be used to complete the assessment of the reference metrics by considering the users' feedback. Finally, the operational concept may be further refined to consider different improvements, including: variable-size safety bubbles for alerting, according to the features of intruder drones (e.g., speed); mitigation of the increase in workload to perform the operational procedure in case of intrusions; and management of additional types of drone intrusions, such as swarm intrusions or cyber drone intrusions.

Author Contributions: Conceptualization, D.P., G.G., A.V., M.S., M.I., P.B., T.D. and E.M.; data curation, D.P., M.S. and M.I.; formal analysis, D.P., G.G., P.B. and T.D.; funding acquisition, M.I., A.V., P.B. and E.M.; methodology, D.P., G.G., P.B. and T.D.; software, D.P., G.G. and M.S.; validation, D.P. and M.S.; visualization, D.P., M.S., P.B. and T.D.; writing—original draft, D.P., G.G., M.S., P.B. and T.D.; writing—review and editing, D.P., G.G., A.V., M.S., M.I., P.B., T.D. and E.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the SESAR Joint Undertaking, grant agreement No 892036, ASPRID project (Airport System PRotection from Intruding Drones).

Data Availability Statement: Publicly available datasets were generated and analyzed in this study related to the configuration and the logs of the real-time simulation exercises. This data can be found here: https://github.com/MaurizioSodano/asprid-file-storage (accessed on 7 April 2023).

Acknowledgments: This research has been started within GARTEUR Aviation Security Group of Responsables.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

Appendix A

This appendix reports the detailed results about the simulation activities for the evaluation of the ASPRID operational concept. The results regard the estimation of the performance metrics for both reference scenarios and solution scenarios. The metrics are estimated for each threat scenario. The following tables (Tables A1–A5) list the detailed results for each metric set (safety, time performance, workload, and capacity).

Table A1. Simulation results: safety metrics of reference scenarios. The following colors are used for SDM values: red for catastrophic (4) value; orange for hazardous (3) value; yellow for major (2) value; green for minor (1) value. Bold values respectively represent the mean and the standard deviation of SDM values.

Threat Scenario	NYI	NOI	NRI	MTYI (s)	MTOI (s)	MTRI (s)	NYI-P	NOI-P	NRI-P	NYI-A	NOI-A	NRI-A	SDM
Threat on 35L—Drone 01	8	3	2	74.38	71.00	1906.50	2	6	0	3	0	2	4
Threat on 35L—Drone 02	12	5	2	772.80	445.78	1011.45	3	9	0	5	0	2	4
Threat on 35L—Drone 03	5	4	1	51.40	1544.00	2047.00	0	5	0	4	0	1	4
Threat on 35R—Drone 01	5	4	1	418.58	1332.18	1760.90	1	4	0	4	0	1	4
Threat on 35R—Drone 02	2	2	1	36.00	1102.00	2188.00	0	2	0	2	0	1	4
Threat on 35R—Drone 03	4	2	1	1076.85	1098.35	2103.70	0	4	0	2	0	1	4
Threat on fingers—Drone 01	10	6	2	1178.08	970.48	41.50	3	7	0	6	0	2	4
Threat parallel to runways—Drone 01	8	0	0	591.35	0.00	0.00	1	7	0	0	0	0	4
Threat parallel to runways—Drone 04	3	0	0	198.67	0.00	0.00	0	3	0	0	0	0	4
Threat on routes—Drone 01	2	0	0	2033.50	0.00	0.00	0	2	0	0	0	0	4
Threat on routes—Drone 03	2	2	0	159.50	1976.50	0.00	0	2	0	2	0	0	4
Threat on tower—Drone 01	11	4	1	1374.64	22.75	12.00	3	8	0	4	0	1	4
Mean Standard Deviation	6.00 3.51	2.67 1.93	0.92 0.76	663.81 611.08	713.59 678.33	922.59 955.23	1.08 1.26	4.92 2.36	0.00 0.00	2.67 1.93	0.00 0.00	0.92 0.76	4.00 0.00

Table A2. Simulation results: safety metrics of solution scenarios. The following colors are used for SDM values: red for catastrophic (4) value; orange for hazardous (3) value; yellow for major (2) value; green for minor (1) value. Bolds values respectively represent the mean and the standard deviation of SDM values.

Threat Scenario	NYI	NOI	NRI	MTYI (s)	MTOI (s)	MTRI (s)	NYI-P	NOI-P	NRI-P	NYI-A	NOI-A	NRI-A	SDM
Threat on 35L—Drone 01	3	2	0	183.00	333.00	0.00	0	3	0	2	0	0	2
Threat on 35L—Drone 02	10	2	0	284.50	263.00	0.00	4	6	0	2	0	0	2
Threat on 35L—Drone 03	3	2	0	347.00	700.00	0.00	0	3	0	2	0	0	2

Threat Scenario	NYI	NOI	NRI	MTYI (s)	MTOI (s)	MTRI (s)	NYI-P	NOI-P	NRI-P	NYI-A	NOI-A	NRI-A	SDM
Threat on 35R—Drone 01	2	1	0	221.00	293.00	0.00	0	2	0	1	0	0	2
Threat on 35R—Drone 02	1	1	0	43.00	410.00	0.00	0	1	0	1	0	0	2
Threat on 35R—Drone 03	3	2	0	139.00	615.50	0.00	1	2	0	2	0	0	2
Threat on fingers—Drone 01	15	7	2	93.87	47.57	41.50	5	10	1	6	0	2	3
Threat parallel to runways—Drone 01	11	0	1	161.19	0.00	188.00	1	10	0	0	0	0	2
Threat parallel to runways—Drone 04	16	0	2	236.23	0.00	249.00	4	12	0	0	0	0	2
Threat on routes—Drone 01	2	0	0	430.45	0.00	0.00	0	2	0	0	0	0	1
Threat on routes—Drone 03	2	2	0	374.50	695.50	0.00	0	2	0	2	0	0	2
Threat on tower—Drone 01	7	2	0	427.29	481.50	0.00	2	5	0	2	0	0	2
Mean Standard Deviation	6.25 5.18	1.75 1.79	0.42 0.76	245.09 123.41	319.92 256.98	39.88 81.64	1.42 1.80	4.83 3.65	0.08 0.28	1.67 1.55	0.00 0.00	0.17 0.55	2.00 0.41

Table A2. Cont.

Table A3. Simulation results: time performance metrics. The following colors are used for TDM values: red for high (2) value; orange for medium (1) value; green for low (0) value. Bold values respectively represent the mean and the standard deviation of TDM values.

	Reference Scenarios								Solution Scenarios					
Threat Scenario	Departures	Takeoffs	NDDF	PDDF (%)	MDDF (s)	WDDF (s)	TDM	Departures	Takeoffs	NDDF	PDDF (%)	MDDF (s)	WDDF (s)	TDM
Threat on 35L—Drone 01	11	2	10	90.91	1328.81	2071.12	2	15	7	8	53.33	400.29	568.57	0
Threat on 35L—Drone 02	9	2	8	88.89	1358.29	2071.12	2	14	6	8	57.14	393.38	739.22	0
Threat on 35L—Drone 03	8	2	7	87.50	1423.03	2071.12	2	17	10	8	47.06	382.44	650.01	0
Threat on 35R—Drone 01	11	4	9	81.82	1085.75	2071.12	1	13	6	6	46.15	225.54	472.10	0
Threat on 35R—Drone 02	8	1	7	87.50	1720.18	2596.10	2	11	4	4	36.36	355.15	612.10	0
Threat on 35R—Drone 03	8	1	7	87.50	1725.79	2596.10	2	17	11	11	64.71	384.96	736.40	0
Threat on fingers—Drone 01	9	2	8	88.89	1381.82	2071.12	2	15	9	7	46.67	336.19	538.22	0
Threat parallel to runways—Drone 01	9	2	8	88.89	1371.04	2071.12	2	13	5	6	46.15	459.20	1111.07	0
Threat parallel to runways—Drone 04	15	8	7	46.67	292.14	546.10	0	19	11	14	73.68	391.65	948.81	0
Threat on routes—Drone 01	9	2	8	88.89	1357.77	2071.12	2	19	12	12	63.16	648.00	1184.22	1
Threat on routes—Drone 03	9	1	8	88.89	1626.39	2596.10	2	17	10	12	70.59	582.62	1161.42	0
Threat on tower—Drone 01	8	1	7	87.50	1718.46	2596.10	2	15	8	7	46.67	331.52	602.12	0
Mean Standard Deviation	9.50 1.94	2.33 1.89	7.83 0.90	84.49 11.59	1365.79 374.57	2119.03 532.33	1.75 0.60	15.42 2.36	8.25 2.52	8.58 2.87	54.31 11.03	407.58 107.83	777.02 245.82	0.08 0.28

Table A4. Simulation results: workload metrics. The following colors are used for WDM values: red for high (2) value; orange for medium (1) value; green for low (0) value. Bold values respectively represent the mean and the standard deviation of WDM values.

Threat Sconario		Reference	e Scenarios		Solution Scenarios				
Threat Scenario	Departures	NCDF	NNCDF	WDM	Departures	NCDF	NNCDF	WDM	
Threat on 35L—Drone 01	11	92	8.36	1	15	71	4.73	0	
Threat on 35L—Drone 02	9	39	4.33	0	14	64	4.57	0	
Threat on 35L—Drone 03	8	36	4.50	0	17	105	6.18	1	
Threat on 35R—Drone 01	11	49	4.45	0	13	56	4.31	0	
Threat on 35R—Drone 02	8	34	4.25	0	11	44	4.00	0	
Threat on 35R—Drone 03	8	34	4.25	0	17	91	5.35	1	
Threat on fingers—Drone 01	9	44	4.89	0	15	85	5.67	1	

Threat Sconario		Reference	e Scenarios		Solution Scenarios				
	Departures	NCDF	NNCDF	WDM	Departures	NCDF	NNCDF	WDM	
Threat parallel to runways—Drone 01	9	38	4.22	0	13	70	5.38	1	
Threat parallel to runways—Drone 04	15	99	6.60	1	19	109	5.74	1	
Threat on routes—Drone 01	9	38	4.22	0	19	104	5.47	1	
Threat on routes—Drone 03	9	37	4.11	0	17	88	5.18	1	
Threat on tower—Drone 01	8	33	4.13	0	15	77	5.13	1	
Mean Standard Deviation	9.50 1.94	47.75 21.83	4.86 1.24	0.17 0.37	15.42 2.36	80.33 19.51	5.14 0.61	0.67 0.47	

Table A4. Cont.

Table A5. Simulation results: capacity metrics.

Threat Scenario	Reference S	cenarios	Solution Scenarios			
Threat Scenario	Departures	DR	Departures	DR		
Threat on 35L—Drone 01	11	1.50	15	4.44		
Threat on 35L—Drone 02	9	0.76	14	4.92		
Threat on 35L—Drone 03	8	0.75	17	5.31		
Threat on 35R—Drone 01	11	1.50	13	6.90		
Threat on 35R—Drone 02	8	0.38	11	6.14		
Threat on 35R—Drone 03	8	0.38	17	5.49		
Threat on fingers—Drone 01	9	0.75	15	6.06		
Threat parallel to runways—Drone 01	9	0.75	13	4.96		
Threat parallel to runways—Drone 04	15	5.45	19	4.13		
Threat on routes—Drone 01	9	0.75	19	5.26		
Threat on routes—Drone 03	9	0.38	17	4.92		
Threat on tower—Drone 01	8	0.38	15	5.63		
Mean	9.50	1.14	15.42	5.35		
Standard Deviation	1.94	1.35	2.36	0.73		

References

- Bouarfa, S.; Blom, H.A.; Curran, R.; HC Everdij, M. Agent-based modeling and simulation of emergent behavior in air transportation. *Complex Adapt. Syst. Model.* 2013, 1, 15. [CrossRef]
- 2. Leonhardt, J.; Macchi, L.; Hollnagel, E.; Kirwan, B. A White Paper on Resilience Engineering for ATM; EUROCONTROL: Brussels, Belgium, 2009. [CrossRef]
- Hollnagel, E. Resilience—The Challenge of the Unstable. In *Resilience Engineering: Concepts and Precepts*, 1st ed.; Hollnagel, E., Woods, D.D., Leveson, N., Eds.; CRC Press: London, UK, 2006; pp. 9–17. [CrossRef]
- Gluchshenko, O.; Foerster, P. Performance Based Approach to Investigate Resilience and Robustness of an ATM System. In Proceedings of the Tenth USA/Europe Air Traffic Management Research and Development Seminar (ATM Seminar 2013), Chicago, IL, USA, 10–13 June 2013.
- Environmental Science Associates, Defining Airport Resilience. Available online: https://esassoc.com/news-and-ideas/2021/0 8/defining-airport-resilience (accessed on 16 January 2023).
- 6. Shelley, A. Drone Registration Will Not Prevent Another Gatwick. SSRN 2019. [CrossRef]
- European Union Aviation Safety Agency (EASA). Drone Incident Management at Aerodromes. Part 1: The Challenge of Unauthorised Drones in the Surroundings of Aerodromes; EASA: Cologne, Germany, March 2021; Available online: https://www.easa.europa.eu/ en/downloads/125405/en (accessed on 16 January 2023).

- European Union Aviation Safety Agency (EASA). The European Plan for Aviation Safety (EPAS 2021–2025), Volume I: Introduction and Strategy; EASA: Cologne, Germany, December 2020; Available online: https://www.easa.europa.eu/en/downloads/123562/en (accessed on 16 January 2023).
- 9. European Union Aviation Safety Agency (EASA). Drone Incident Management at Aerodromes. Part 2: Guidance and Recommendations; EASA: Cologne, Germany, March 2021.
- 10. European Union Aviation Safety Agency (EASA). Drone Incident Management at Aerodromes. Part 3: Resources and Practical Tools; EASA: Cologne, Germany, March 2021.
- 11. ASPRID (Airport System PRotection from Intruding Drones). Available online: https://www.asprid.eu/ (accessed on 16 January 2023).
- 12. ASPRID Presentation in Drones Beyond 2022. Available online: https://www.youtube.com/watch?v=7bebCiZtNYs (accessed on 16 January 2023).
- Pascarella, D.; Gigante, G.; Nebula, F.; Vozella, A.; Redondo de la Mata, E.; Jiménez Roncero, F.J.; Martinavarro, E. Historical Data Analysis and Modelling for Drone Intrusions in Airports. In Proceedings of the 2021 13th International Conference on Electronics, Computers and Artificial Intelligence (ECAI), Pitesti, Romania, 1–3 July 2021; pp. 1–8. [CrossRef]
- 14. Pascarella, D.; Gigante, G.; Vozella, A.; Bieber, P.; Dubot, T.; Martinavarro, E.; Barraco, G.; Li Calzi, G. A Methodological Framework for the Risk Assessment of Drone Intrusions in Airports. *Aerospace* **2022**, *9*, 747. [CrossRef]
- 15. Guo, J.; Zhu, X.; Liu, C.; Ge, S. Resilience Modeling Method of Airport Network Affected by Global Public Health Events. *Math. Probl. Eng.* **2021**, 2021, 6622031. [CrossRef]
- 16. Cook, A.; Blom, H.A.P.; Lillo, F.; Mantegna, R.N.; Miccichè, S.; Rivas, D.; Vázquez, R.; Zanin, M. Applying complexity science to air traffic management. *J. Air Transp. Manag.* **2015**, *42*, 149–158. [CrossRef]
- 17. Resilience2050. Available online: https://resilience2050.org (accessed on 16 January 2023).
- Gluchshenko, O.; Foerster, P. Setting the Basis—Defining Resilience in the Organization of ATM; V1.0; Seventh Framework Programme of the European Union: Brussels, Belgium, 30 August 2013; Available online: https://s3.eu-central-1.amazonaws.com/innaxiscomm/RESILIENCE2050/D1.3+Resilience2050.pdf (accessed on 16 January 2023).
- The Innaxis Foundation and Research Institute. *Resilience Multilayer Representationz*; Issue 1 Draft 1; Seventh Framework Programme of the European Union: Brussels, Belgium, 30 September 2013; Available online: https://s3.eu-central-1.amazonaws. com/innaxis-comm/RESILIENCE2050/D3.1+Resilience2050.pdf (accessed on 16 January 2023).
- Filippone, E.; Gargiulo, F.; Errico, A.; Di Vito, V.; Pascarella, D. Resilience management problem in ATM systems as a shortest path problem. J. Air Transp. Manag. 2016, 56 Pt A, 57–65. [CrossRef]
- Pascarella, D.; Gargiulo, F.; Errico, A.; Filippone, E. An Analytical Approach for Optimal Resilience Management in Future ATM Systems. In *Intelligent Distributed Computing IX*; Studies in Computational Intelligence, Novais, P., Camacho, D., Analide, C., El Fallah Seghrouchni, A., Badica, C., Eds.; Springer: Cham, Switzerland, 2016; Volume 616, pp. 415–425. [CrossRef]
- 22. European Organisation for Civil Aviation Equipment (EUROCAE). ED-286. Operational Services and Environment Definition for Counter-UAS in Controlled Airspace; EUROCAE: Saint-Denis, France, March 2021.
- 23. Airports Council International (ACI) Europe. Drones in the Airport Environment: Concept of Operations & Industry Guidance; ACI Europe: Brussels, Belgium, April 2020.
- 24. Pyrgies, J. The UAVs threat to airport security: Risk analysis and mitigation. J. Airl. Airpt. Manag. 2019, 9, 63–96. [CrossRef]
- 25. Lykou, G.; Moustakas, D.; Gritzalis, D. Defending Airports from UAS: A Survey on Cyber-Attacks and Counter-Drone Sensing Technologies. *Sensors* **2020**, *20*, 3537. [CrossRef]
- 26. Castrillo, V.U.; Manco, A.; Pascarella, D.; Gigante, G. A Review of Counter-UAS Technologies for Cooperative Defensive Teams of Drones. *Drones* **2022**, *6*, 65. [CrossRef]
- Kang, H.; Joung, J.; Kim, J.; Kang, J.; Cho, Y.S. Protect Your Sky: A Survey of Counter Unmanned Aerial Vehicle Systems. *IEEE Access* 2020, *8*, 168671–168710. [CrossRef]
- Guide to the Systems Engineering Body of Knowledge (SEBoK)—Operational Concept (Glossary). Available online: https://www.sebokwiki.org/wiki/Operational_Concept_(glossary) (accessed on 16 January 2023).
- Object Management Group (OMG). Business Process Model and Notation (BPMN); formal/2013-12-09, 2.0.2; OBM: Needham, MA, USA, December 2013; Available online: https://www.omg.org/spec/BPMN/2.0.2/PDF (accessed on 16 January 2023).
- Manfredi, G.; Jestin, Y. An Introduction to Fast Time Simulations for RPAS Collision Avoidance System Evaluation. In Proceedings
 of the ICRAT 2018, 8th International Conference for Research in Air Transportation, Barcelona, Spain, 26–29 June 2018.
- 31. ASPRID Consortium. *DEMO Software for Validation*; 01.00.01; SESAR Joint Undertaking: Brussels, Belgium, 4 February 2022; Available online: https://www.asprid.eu/wp-content/uploads/2022/05/D3.3.pdf (accessed on 16 January 2023).
- 32. eArts (En-Route Aerodrome Real Time Simulator). Available online: https://www.soulsoftware.it/html/en/earts-1 (accessed on 16 January 2023).
- 33. ASPRID Demo Web App. Available online: https://aspriddemo.web.app (accessed on 16 January 2023).
- 34. Le Fevre, M.; Gölz, B.; Flohr, R.; Stelkens-Kobsch, T.; Verhoogt, T. *SecRAM 2.0 Security Risk Assessment Methodology for SESAR 2020*; 02.00.00; SESAR Joint Undertaking: Brussels, Belgium, 25 September 2017.

- 35. Cordero, J.M.; García-Ovies, I.; Verdonk Gallego, C.; Sánchez-Palomo Bermúdez, L.; Bujor, A.; Calvo Fernandez, E.; Carbo, L.; Deransy, R.; Kanstein, A.; Lacatus, R.; et al. *PJ19.04: Performance Framework* (2019); 01.00.01; SESAR Joint Undertaking: Brussels, Belgium, 30 November 2019.
- ASPRID Gaming Exercise Results. Available online: https://www.asprid.eu/wp-content/uploads/2022/09/220913-ASPRID-Gaming-Exercise-results-ENG.pdf (accessed on 7 April 2023).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.